Author’s response to reviews

Title: The effects of add-on self-care education on quality of life and fatigue in gastrointestinal cancer patients undergoing chemotherapy

Authors:

Jun Xie (x-j1127@163.com)
Tingli Zhu (zhutingli_wuxi@163.com)
Qun Lu (1662935168@qq.com)
Xiaomin Xu (xuxm@wuxiph.com)
Yinghua Cai (caiyh@wuxiph.com)
Zhenghong Xu (xuzh@wuxiph.com)

Version: 3 Date: 13 Dec 2019

Author’s response to reviews:

Reviewer reports:

Johanna Hok Nordberg (Reviewer 2): Thank you for the revised manuscript! I think your revisions improved the clarity of the presentation. I have only one comment:

Page 4, row 22: Your revised suggestion of this sentence is better than before but I think one exchange of words may clarify your intention: Do you mean "self-care" while you write "health care"? ("... availability and accessibility of health care...")?

Response:

Yes, it should be “self-care” and is corrected as so.